Author’s response to reviews

Title: BRIP-1 germline mutation and its role in colon cancer: Presentation of two case reports and review of literature

Authors:

Mir Ali (dirmirali@yahoo.com)
Celia Delozier (cdelozier@communitymedical.org)
Uzair Chaudhary (uchaudhary@fresno.ucsf.edu)

Version: 3 Date: 19 Apr 2019

Reviewer comments and response.

1. Please remove the names and affiliations of authors from the title in your manuscript’s submission data.

Response- I have removed the names and affiliations of authors from the title in manuscript's submission data.

2. Please include a title page as the first page of your manuscript where you list the full names, institutional addresses and the email address for the corresponding author. The corresponding author should still be indicated.

Response- I was a little confused as to how to attach the title page. Based on the BMC medical genetics submission guidelines, I have separated the title on single page in the manuscript along with author names, affiliation, address and corresponding author email. The rest of the article is continued from the next page. Hopefully this meets the format that is being requested.